Author's response to reviews

Title: Antibodies to group A streptococcal virulence factors, SIC and DRS, increase predilection to GAS pyoderma.

Authors:

Mohan G Karmarkar (mohankarmarkar@kem.edu)
Gouri P Hule (gaurihule@gmail.com)
Ainslie Cameron (a.calcutt@griffith.edu.au)
Preeti R Mehta (preetimehta@kem.edu)
Uday S Khopkar (drkhopkar@gmail.com)
Niwrtti K Hase (nk_hase@hotmail.com)
Kadaba S Sriprakash (sris@qimr.edu.au)

Version: 2
Date: 5 September 2014

Author's response to reviews:

Dear Editor,

Thank you for pointing out some of the formatting issues in our manuscript. As suggested, the following actions were taken.

1. statement on competing interests included.
2. Authors’ contributions included.
3. Listed email addresses in the title page for all authors.
4. Abstract is semi-structured.
5. Conclusions added.
6. Line numbers and page numbers inserted.